Author's response to reviews

Title: Severe anemia due to pharyngeal leech infestation; A case report from Ethiopia

Authors:

Nebiyu Shitaye (handworklove@gmail.com)

segenet shibabaw (segenetlove@gmail.com)

Version: 3 Date: 19 Sep 2017

Author’s response to reviews:

Dear Editor

1. Regarding the significant text overlap between our case report and other publications;

As you already said the software that you are using to detect these overlaps is not perfect and if possible it is good if someone check these overlaps manually. For example let me show you how much it is inappropriate

A. Bahir Dar University College of Medicine and Health Science, Felege Hiwot referral Hospital, Department of public health; Bahir Dar Ethiopia; Email: - segenetlove@gmail.com

So the above statement is just the full address of the coauthor (Segenet Shibabaw) but the software detects as if we copied it from “Abrham Jemberie Temach, Gedefaw Abeje Fekadu, Anemaw Asrat Achamyeleh. "Educational status as determinant of men’s knowledge about vasectomy in Dangila town administration, Amhara region, Northwest Ethiopia", Reproductive Health, 2017” Crossref

So two or more people may have the same address as it happens in the above scenario and obviously they will write the same thing and the software may detect that. But we cannot correct this because we cannot change our address.

B. In the discussion part we are comparing our report with others publications in the literature, so we bring some bodies work but the references are there. Still the comment is accepted and the statement is reformulated without deviating from the main concept.

C. If you go to the reference, the whole reference is detected as overlap from different sources. It is obvious that the name of the reference is somewhere in the literature, that is why we put it as a reference. And as you well know we can’t reformulate or modify the name of the references. But the software continued to detect it.

2. Figure legends given a separate title next to the reference….done
3. Figure submission as a separate and single file….done

4. Declarations….addressed